Author’s response to reviews

Title: Dual energy CT findings in gout with rapid kilovoltage-switching source with gemstone scintillator detector

Authors:

Elin Svensson (elin.g.svensson@vgregion.se)
Ylva Aurell (ylva.aurell@vgregion.se)
Lennart Jacobsson (lennart.jacobsson@gu.se)
Anton Landgren (anton.landgren@rheuma.gu.se)
Valgerdur Sigurdardottir (Valgerdur.Sigurdardottir@ltdalarna.se)
Mats Dehlin (mats.dehlin@vgregion.se)

Version: 2 Date: 21 Oct 2019

Author’s response to reviews:

1. Title page

Please indicate the corresponding author on the title page.

Mats Dehlin is indicated as the corresponding author

2. Declarations

Please move the acknowledgments section into your declarations.

Acknowledgements have been moved into Declaration section
3. Consent for publication

We note that you have not included a ‘Consent for publication’ section in Declarations. If identifying images or other personal or clinical details of participants are presented that compromise anonymity, a statement of consent to publish from the patient should be included. This section must be included even if it is not applicable to your manuscript. If consent to publish is not applicable to your manuscript please write ‘Not Applicable’ in this section.

Consent for publication have been included in the Declaration section

4. Funding

In the Funding section, please also describe the role of the funding body in the design of the study and collection, analysis, and interpretation of data and in writing the manuscript.

The role of the funding has been added.

5. Identifying information

Currently, the age and gender information present in your figure legends, in combination with other identifiers, may compromise patient/participant anonymity.

As you do not have consent to publish, please amend the file to address this issue, by replacing this information with "a patient".

Figure legend for Figure have been modified to "a patient".
6. Clean manuscript

At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files. Please ensure that all figures, tables and additional/supplementary files are cited within the text.

The manuscript has been cleaned and all figures and tables are cited within the text.

Figure legend for Figure 1 has been written in the description of Figure 1 in the Attach files section

Regards Mats Dehlin